Author's response to reviews

Title: Skin-impedance in Fabry Disease: A prospective, controlled, non-randomized clinical study

Authors:

Surya Gupta (Surya.Gupta@tuhs.temple.edu)
Markus Ries (mries@shire.com)
Gary J Murray (murrayg@ninds.nih.gov)
Jane M Quirk (quirkj@ninds.nih.gov)
Roscoe O Brady (brayr@ninds.nih.gov)
Jeffrey R Lidicker (jeffrey.lidicker@uhs.temple.edu)
Raphael Schiffmann (raphael.schiffman@baylorhealth.edu)
David F Moore (David.F.Moore@amedd.army.mil)

Version: 4 Date: 15 August 2008

Author's response to reviews:

August 15, 2008

To the Editor of Biomed Central Neurology

Re: 2003602465179258 Resubmission 2 - Skin-impedance in Fabru disease: A prospective, controlled, non-randomized clinical study

Dear Editor,

We made the two changes in the Discussion requested by Dr. Robert Steiner. They are highlighted in yellow.

As we have addressed all the reviewers' concerns, we hope this manuscript is now acceptable for publication in Biomed Central.

Sincerely,

Raphael Schiffmann, M.D.